Author's response to reviews

Title:	Screening pregna	nt women fo	r suicidal	behavior is	n electronic	medical records:	diagnostic
codes	vs. clinical notes	processed by	natural la	nguage pro	cessing		

Δ	11	1	h	Λ	rs	
$\overline{}$	•		•	.,		_

Qiu-Yue Zhong (qyzhong@mail.harvard.edu)

Elizabeth Karlson (EKARLSON@partners.org)

Bizu Gelaye (bgelaye@hsph.harvard.edu)

Sean Finan (Sean.Finan@childrens.harvard.edu)

Paul Avillach (Paul_Avillach@hms.harvard.edu)

Jordan Smoller (jsmoller@hms.harvard.edu)

Tianxi Cai (tcai@hsph.harvard.edu)

Michelle Williams (mawilliams@hsph.harvard.edu)

Version: 1 Date: 03 Nov 2017

Author's response to reviews:

Dear Catherine,

Thank you for your email.

We have carried out the corrections as indicated in the email.

Best,

Qiu-Yue